Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
Feb. 16, 2011
Injector    Energy 61.76 MeV
· (2/15) Beam studies: matching the injector to improve Hall C beam quality. After the match all three beam were different. The injector did not reproduce. 
· (2/14) Capture water skid tripped off once. We have the procedure to reset it.

· (2/14) The laser spot was moved to boost the QE.

· (2/11) The injector was optimized to improve the beam quality (mainly for Hall C halo and Compton counts).

· (2/10) The vertical polarization was backed out and a spin dance was performed. The HWien was set to -62.4 degrees. 

· (2/10) The 45MeV spectrometer power line was re-connected. The magnet is still off.
· (2/8) The beam polarization was set to vertical for Hall C. 
· (2/4) It was found that cavity 0L03-2 was the root cause of the injector instability. The gradient of the cavity was reduced by 2MeV and we were able to send 150uA to Hall C. The beam image on 1A SLM still looks abnormal. We will investigate at a later time since both halls are happy with beam and we have lost a lot of time due to the investigation of the injector instability. 
· (2/3) New optics was loaded in the injector and beam appears to be better afterward. 

· (2/3) The 45MeV spectrometer dipole was degaussed and turned off during the troubleshooting of the injector instability and is still off.

Accelerator   Linac Energy 549.00 MeV
· Main machine has been stable.
· (2/15) Pass 1 separator power dropped to zero and EES repaired it.

· (2/12) 2L05 waveguide pressure line leak and was fixed.  Cavity 2L05-2 control module was replaced.
· (2/12) Cavity 0L03-2 became unstable: the gradient was lowered to 3MeV. There were a lot if 2T09 BLM trips. Lowering the gradient of cavity 0L03-2 and steering 2T and 4T reduced the BLM trips.

· (2/10) Pass 1 separator was repaired and is in use. Pass 2 separator is for pass 2 now.

· (2/10) The DAC for 1L02 and 1L03 heater control was replaced.

· When to send beam after a vacuum event: whenever the beam line valves can be opened we can send beam.

· The optics has been stable.
· (2/8) There was water leak on 1A12 girder. The leak was fixed.
· (2/8) SL dropped to restricted access. No obvious cause indicated on the PSS screen.
· (2/7) Arc9A box supply earth leakage fault (~ 3 hrs.) that might be caused by water leak. 
Hall A---Pass 2, E08 - 010
· Beam does not go through the Compton for this run.
· New Hall A Moller procedure.

· Will go to pass 3 Thursday.
· (2/12) Started to take low current beam (< 5uA).
· (2/9) Started to hot check out this week and is planning to take beam tonight.
Hall B ---Pass 5, TPE
· Will go to pass 2 Thursday for a couple of shifts.

· The beam is good.
· The beam is fine. 

· (2/8) The hall reported that its beam positions were not stable with current (1nA).
Hall C--- Pass 1, QWeak 
· The beam may not be as good as before the injector match.
· The hall has vacuum issues in the dump area. It is not beam related.

· The hall is happy with its beam. Halo rates are good. We need to fine tune the Compton. 
